Author's response to reviews

Title: Atraumatic patellar prosthesis dislocation with patellar tendon injury following a total knee arthroplasty: a case report

Authors:

Vinay Kumar Singh (orthopaediks@yahoo.co.uk)
Pankaj Kumar Singh (we.publish@googlemail.com)
Yashwant Singh (drvashwant2006@yahoo.com)
Alka Singh (alka.singh@bht.nhs.net)
Sadaf Javed (sadafjaved@doctors.org.uk)
Murad Abdunabi (abdunabi@web.de)

Version: 3 Date: 15 December 2009

Author's response to reviews:

Dear Sir,

May thanks for the opportunity to submit revision. I have made the necessary changes and I hope it meets the requirement of honourable editorial board and the manuscript may accepted for publication. If there are any minor revisons authors do not have any problems if editorial team can do minor changes.

Following are the details of the changes

1. All the authros names have been added as in submission system
2. Patients demographic details have been added in abstract.
3. Note: All authors have read and approved the final manuscript. This note has been added on authors contributions.

Many thanks
Mr Singh
Many thanks
Mr Singh